**Springer Series in Reliability Engineering** 

Xiao-Sheng Si Zheng-Xin Zhang Chang-Hua Hu

# Data-Driven Remaining Useful Life Prognosis Techniques

Stochastic Models, Methods and Applications



# **Springer Series in Reliability Engineering**

Series editor

Hoang Pham, Piscataway, USA

More information about this series at http://www.springer.com/series/6917

Xiao-Sheng Si · Zheng-Xin Zhang Chang-Hua Hu

# Data-Driven Remaining Useful Life Prognosis Techniques

Stochastic Models, Methods and Applications





Xiao-Sheng Si Department of Automation Xi'an Institute of High-Technology Xi'an, Shaanxi China

Zheng-Xin Zhang Department of Automation Xi'an Institute of High-Technology Xi'an, Shaanxi China Chang-Hua Hu Department of Automation Xi'an Institute of High-Technology Xi'an, Shaanxi China

ISSN 1614-7839 ISSN 2196-999X (electronic) Springer Series in Reliability Engineering ISBN 978-3-662-54028-2 ISBN 978-3-662-54030-5 (eBook) DOI 10.1007/978-3-662-54030-5

Jointly published with National Defense Industry Press, Beijing, China

Library of Congress Control Number: 2016961678

© National Defense Industry Press and Springer-Verlag GmbH Germany 2017

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Printed on acid-free paper

This Springer imprint is published by Springer Nature The registered company is Springer-Verlag GmbH Germany The registered company address is: Heidelberger Platz 3, 14197 Berlin, Germany

#### Preface

The remaining useful life (RUL) of a system is defined as the length from the current time to the end of the useful life. The concept of the RUL has been widely used in operational research, reliability, and statistics literature with important applications in other fields such as materials science, biostatistics, and econometrics. However, there are many definitions as what is regarded as the useful life. In 'Businessdictionary.com,' it defines the useful life "the period during which an asset or property is expected to be usable for the purpose it was acquired'. However, in accounting, it is defined as 'the expected period of time during which a depreciating asset will be productive." The keyword here is 'usable' or 'productive' which is again upon individual explanations. Clearly the definition of the useful life depends on the context and operational characteristics. In this book we will assume that the definition of the useful life is known to the owner of the asset and the main interest is to investigate the modeling methods for RUL estimation given condition and health monitoring information.

In conventional data-based approaches, estimating the RUL is achieved by evaluating the conditional lifetime distribution given that a system has survived up to a specific time. The obtained RUL distributions from these approaches are generally based on the life characteristics of a population of identical systems and lifetime data are required. However, such data are scarce in reality or even nonexistent at all for systems which are costly or time-consuming to collect the life data. With the advances in CM technologies, degradation data can be obtained from routine CM as feasible and low-cost alternatives to estimate the RUL. These data are usually correlated with the underlying physical degradation process. If they are properly modeled, degradation data can be used to predict unexpected failures and accurately estimate the lifetime of gradually degraded systems. In many situations, such as the drift degradation of an inertial navigation system used in the aerospace industry, it is natural to view the failure event of interest as the result of a stochastic degradation process crossing a threshold level, i.e., to model the hitting time of the degradation as a time-dependent stochastic process. On the other hand, dynamic environments induce changes in the physics of failure.

RUL prognosis is one of the key factors in condition-based maintenance (CBM), and prognostics and health management. It is critically important to assess the RUL of an asset while in use since it has impacts on the planning of maintenance activities, spare parts provision, operational performance, and profitability of the owner of an asset. RUL estimation has also an important role in the management of product reuse and recycle which has strategic impacts on energy consumption, raw material use, pollution, and landfill. The reused products must have sufficient long lives left among others to be able to be reused. This puts the importance of the estimation of RUL beyond CBM and prognostics and health management because of the green issues associated. As a consequence, developing RUL prognosis methods is much desired for health management of degrading systems to prevent sudden failure and reduce the safety risk. In the past four decades, valuable contributions to prognostics in reliability field have been made. This book is intended to summarize the research results studied mainly by the authors in the past decade.

This book introduces the main ideas of data-driven remaining useful life prognosis techniques, with an emphasis on stochastic models, methods, and applications. It gives a thorough survey of new methods that have been developed in the recent years and demonstrates them with examples. To the knowledge of the authors, all major aspects of RUL prognosis are treated for the first time in a single book from a common viewpoint. With the presentation of RUL prognosis methods for degrading systems, the book provides novel materials that have not yet been described in monographs or textbooks.

This monograph consists of four parts:

- Part I: Introduction, Degradation Data Acquisition and Evaluation. Advances in data-driven RUL prognosis techniques are reviewed. As fundamental issues for data-driven RUL prognosis, methods of how to acquiring the degradation data and how to evaluate the usability of the acquired data are presented.
- Part II: Prognostic Techniques for Linear Degrading Systems. Methods for adaptive RUL prognosis, exact RUL prognosis solution, RUL prognosis with multiple kinds of variability for linear degrading systems are presented and the methods are demonstrated by case studies.
- Part III: Prognostic Techniques for Nonlinear Degrading Systems. Methods for nonlinear degradation modeling, adaptive RUL prognosis, nonlinear RUL prognosis under multiple sources of variability, residual storage life prognosis with switching systems for nonlinear degrading systems are presented and the methods are demonstrated by case studies.
- Part IV: Applications of Prognostic Information. This part discusses the applications of prognostic information such as mission reliability estimation, condition-based replacement, spare parts forecasting, and joint optimization of spare part ordering and replacement.

As each of the models used requires its own mathematical background and the methods based on these models follow different lines of thinking, the book cannot present the methods for all details. The aim is to give the readers a broad view of the

field and provide them with bibliographical notes for further reading. A further reason for the different depth with which the chapters tackle the RUL prognosis problems is given by the status of research. In the introductory parts of all chapters, the problems to be solved are posed in a framework that is familiar to practicing engineers. They describe the new ideas and concepts of RUL prognosis in an intuitive way, before these ideas are brought into a strict mathematical form. Examples illustrate the applicability of the methods. Bibliographical notes at the end of each chapter point to the origins of the presented ideas and the current research lines. The evaluation of the methods and the application studies should help the readers to assess the available methods and the limits of the present knowledge about RUL prognosis with respect to their particular field of application.

Together with four parts, the book is composed of 16 chapters. Chapter 1 is devoted to an introduction to advances in data-driven RUL prognosis techniques. Chapter 2 considers the problem of planning repeated degradation test for degrading products with three-source variability. In Chap. 3, the attention is paid to specifying measurement errors for required lifetime estimation performance so as to evaluate the data usability. A linear degradation model with a recursive filter algorithm and Bayesian updating is presented to estimate the PDF of the RUL in Chap. 4. Chapter 5 derives the exact and closed-form solution of RUL prognosis for linear degrading systems. Chapter 6 presents a Wiener-process-based degradation modeling framework for RUL estimation with three-source variability. In Chap. 7, a diffusion process-based model was presented to characterize the dynamics and nonlinearity of degradation processes, and the corresponding RUL distribution is formulated. The results in Chap. 7 are further extended to an age- and state-dependent case in Chap. 8. In Chap. 9, an adaptive and nonlinear prognostic model is presented to estimate the RUL using the history of the observed data to date. Chapter 10 develops a real-time RUL estimation method based on a state space model considering that the degradation process is hidden and nonlinear. Chapter 11 presents a general nonlinear diffusion process-based model to estimate the RUL with the temporal variability, unit-to-unit variability, and measurement variability. In Chap. 12, the problem of predicting RSL for a class of systems with operation state switches is concerned. Chapter 13 applies the prognostic information to reliability estimation of phased-mission systems. In Chap. 14, a real-time variable cost-based maintenance model is presented based on nonlinear prognostic information. Chapter 15 presents an adaptive spare parts demand forecasting method based on degradation modeling of the CM data. In Chap. 16, a new sequential maintenance and inventory model is developed to consider the effects of both expectation of the maintenance cost and its variability under prognostic information.

In preparing the book, efforts have been made to maintain a balance between the required theoretical and mathematical rigor in the exposition of the methods and the clarity in the illustration of the numerical examples and practical applications. For this reason, this book can serve well as a reference to both reliability and risk analysis researchers and engineers. Furthermore, sufficient references leading to further studies are cited at the end of each chapter. This book will serve as a textbook and

reference book for graduate students and researchers in reliability and maintenance. Although the book is self-explanatory, a standard background in probability theory, mathematical statistics, and stochastic processes is recommended.

Finally, we wish to thank Profs. Wenbin Wang, Donghua Zhou, and Michael Pecht for their cooperation and valuable discussions. In addition, it is with sincere appreciation that we thank the support by National Nature Science Foundation of China under Grant 61174030, 61374126, 61473094, 61573076, 61573366, and the NSF of Shaanxi Province of China under grant 2015JQ6235.

Xi'an, China July 2016 Xiao-Sheng Si Zheng-Xin Zhang Chang-Hua Hu

#### Contents

# Part I Introduction, Degradation Data Acquisition and Evaluation

1	Advances in Data-Driven RUL Prognosis Techniques			3		
	1.1	Introdu	action	3		
	1.2	Metho	ds Considering Unit-to-Unit Variability	5		
		1.2.1	Random Coefficients Regression Models	6		
		1.2.2	Stochastic Process Models with Random			
			Coefficients	6		
	1.3	Metho	ds Considering Impact of Heterogeneity			
		in Woi	cking Environment	8		
		1.3.1	Methods Based on Stochastic Filtering	8		
		1.3.2	Multi-stage Degradation Models	9		
		1.3.3	Covariate Hazards Model	10		
		1.3.4	Degradation Models Involving Random Shocks	11		
	1.4	Metho	ds Considering the Impact of Tasks and Workloads	13		
		1.4.1	Degradation Modeling for Systems with Dynamic			
			Workloads	13		
		1.4.2	Degradation Modeling for System			
			with Maintenances	14		
	1.5	Future	Research Directions	15		
	Refe	rences		17		
2	Planning Repeated Degradation Testing					
	for I	for Degrading Products				
	2.1	Introdu	iction	23		
	2.2	Degrad	lation Modeling with Three-Source Variability	25		
	2.3	Parame	eter Estimation and Information Matrix	27		

	2.4	Estimating the Degradation Distribution and Lifetime Distribution	28
		2.4.1 The Quantiles of Degradation Distribution	
		and Its Variance	28
		2.4.2 The Lifetime Distribution	29
	2.5	Degradation Test Planning	32
	2.6	An Illustrative Example	33
	Refere	ences.	36
3	Specif	fying Measurement Errors for Required Lifetime	
	Estim	nation Performance	39
	3.1	Introduction	39
	3.2	Properties of the WPDM	42
	3.3	Properties of the WPDM with the ME	43
	3.4	Permissible ME Parameters for Lifetime Estimation	45
		3.4.1 Performance Measures to Quantify the Difference	
		in Lifetime Estimation with Versus Without	
		the ME	45
		3.4.2 Permissible ME Parameters Using the Relative	
		Increase Ratio of the CV	46
		3.4.3 Permissible ME Parameters Using the Relative	
		Increase Ratio of the Variance	49
	35	Effect of Lifetime Estimation with or Without ME	.,
	0.0	on an Age-Based Replacement Decision	50
	36	Experimental Studies	52
	5.0	3.6.1 A Numerical Illustration	52
		3.6.2 The Case Study	55
	Annei	ndiv	62
	Defer	nuix	67
	Kelen		07
Par	t II F	Prognostic Techniques for Linear Degrading Systems	
4	An A	daptive Remaining Useful Life Estimation Approach	
	with a	a Recursive Filter	73
	4.1	Introduction	73
	4.2	Wiener-Process-Based Degradation Modeling and RUL	
		Estimation	76
		4.2.1 An Outline of Wiener-Process-Based Degradation	
		Model for Lifetime Analysis	76
		4.2.2 Wiener-Process-Based Degradation Modeling	79
		4.2.3 Real-Time Updating of the RUL Distribution	81
	4.3	Parameter Estimation	84
		4.3.1 EM Algorithm	84
		4.3.2 The Implementation of EM Algorithm	
		for the Proposed Model	86

#### Contents

		4.3.3	Convergence Analysis of Adaptive Model	
			Parameter Estimation Algorithm	90
	4.4	A Prac	tical Case Study	92
		4.4.1	Problem Description	92
		4.4.2	The Implementation of Our Model for RUL	
			Estimation of the INS	94
		4.4.3	Comparative Studies	96
	Refe	rences		100
5	An F	Exact an	d Closed-Form Solution to Degradation	
	Path	-Depend	ent RUL Estimation	103
	5.1	Introdu	iction	103
	5.2	A Deg	radation Path-Dependent Approach for Adaptive	
		RUL E	Estimation	107
		5.2.1	A General Description of Stochastic Process	
			Based Degradation Models	107
		5.2.2	A Degradation Path-Dependent Approach	
			for Adaptive RUL Estimation via Real-Time	
			CM Data	108
	5.3	Linear	Model	110
	5.4	Expon	ential Model	123
	5.5	Experi	mental Studies	130
		5.5.1	Numerical Example	131
		5.5.2	A Practical Case Study of the Developed	
			Approach in Condition-Based Replacement.	136
	Refe	rences		141
6	Estir	nating <b>R</b>	RUL with Three-Source Variability	
	in D	egradati	on Modeling	143
	6.1	Introdu	iction	143
		6.1.1	Motivation	143
		6.1.2	Related Works	145
		6.1.3	Main Works of This Chapter	146
	6.2	Descri	ption of Degradation Modeling with Three-Source	
		Variab	ility for RUL Estimation.	147
	6.3	RUL E	Estimation with Three-Source Variability	149
		6.3.1	RUL Estimation with Temporal Variability	
			and Unit-to-Unit Variability.	149
		6.3.2	RUL Estimation with Temporal Variability	
			and Uncertain Measurements	154
		6.3.3	RUL Estimation with Three-Source Variability	158
	6.4	Parame	eter Estimation	167

	6.5 Refer	Experin 6.5.1 6.5.2 6.5.3 rences	nental Studies Problem Description Comparisons for Model Fitting Comparisons for the Estimated RUL	170 171 174 175 178
Part	t III	Prognos	stic Techniques for Nonlinear Degrading Systems	
7	RUL	Estimat	tion Based on a Nonlinear Diffusion Degradation	
	Proc	ess		183
	7.1	Introdu	ction	183
	7.2	Literati	re Review	186
	7.3	Motiva	ting Examples and RUL Modeling Principle	187
	7.4	Lifetim	e Distribution and Parameter Estimation	
		of the 1	Proposed Degradation Model	191
		7.4.1	Derivation of the Lifetime Distribution	191
		7.4.2	Lifetime Distribution Under Random Effects	198
		7.4.3	The Distribution of the RUL Estimation	200
	7.5	Parame	eters Estimation.	202
	7.6	Examp	les of the Applications of the Models	205
		7.6.1	Laser Data	206
		7.6.2	Drift Degradation Data of INS	208
		7.6.3	Fatigue Crack Data of 2017-T4	211
	Refer	ences		213
8	Prog	nostics f	or Age- and State-Dependent Nonlinear	
-	Degr	ading Sy	vstems	217
	8.1	Introdu	ction	217
	8.2	Proble	n Formulation	219

	8.3	RUL E	Estimation by Degradation Modeling	221
	8.4	Model	Parameter Estimation Framework	226
	8.5	An Illu	strative Example	228
		8.5.1	Degradation Model and Lifetime Estimation	228
		8.5.2	Parameters Estimation	230
		8.5.3	Verifying the Accuracy of the Proposed Method	231
	8.6	Case S	Study	232
	Refe	rences	· · · · · · · · · · · · · · · · · · ·	243
9	Adaj	otive Pro	ognostic Approach via Nonlinear Degradation	
	Mod	eling	······	247
	9.1	Introdu	action	247
	9.2	Nonlin	ear Model Description and RUL Estimation.	250
		9.2.1	Modeling Description	250
		9.2.2	Derivation of the RUL Distribution	251
		0.0.0		050

	9.3	Adaptiv	ve Parameter Estimation	256
	9.4	An Illu	strative Example	260
	9.5	Numeri	cal Example and Case Study	261
		9.5.1	Numerical Example	261
		9.5.2	Lithium-Ion Battery Life Prognosis	265
	Refer	ences		268
10	Prog	nostics f	or Hidden and Age-Dependent Nonlinear	
	Degr	ading Sy	vstems	273
	10.1	Introdu	ction	273
		10.1.1	Motivation	273
		10.1.2	Related Works	274
		10.1.3	Main Works of This Chapter	276
	10.2	Problem	n Formulation and RUL Estimation	277
		10.2.1	Problem Formulation	277
		10.2.2	RUL Estimation.	279
		10.2.3	Comparative Discussions	284
	10.3	Parame	ter Estimation	287
	10.4	Illustrat	tive Examples	291
		10.4.1	The Derivation of the RUL for Three Cases	291
		10.4.2	The Derivation of Parameter Estimation	
			Algorithm for Three Cases	293
	10.5	Simulat	tion Study	297
	10.6	Case St	tudy	305
		10.6.1	The Data and State-Space-Based Degradation	
			Model	305
		10.6.2	Results and Discussions.	307
	Refer	ences		309
11	Prog	nostics f	or Nonlinear Degrading Systems	
	with	Three-Se	ource Variability	313
	11.1	Introdu	ction	313
	11.2	Nonline	ear Prognostic Model Description	315
	11.3	RUL E	stimate Method with Three-Source Variability	317
		11.3.1	RUL Estimate Only with the Temporal Variability	317
		11.3.2	RUL Estimate with the Temporal Variability	
			and the Unit-to-Unit Variability	318
		11.3.3	RUL Estimate with the Temporal Variability	
			and the Measurement Variability	321
		11.3.4	RUL Estimate with Three-Source Variability	324
		11.3.5	Parameter Estimation	328

Contents
----------

11.4.1       Simulation Study       329         11.4.2       Case Study       332         References       335 <b>12 RSL Prediction Approach for Systems with Operation State</b> Switches       337         12.1       Introduction       337         12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Operation Process       344         12.5       Model Formulation of the System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the       Degradation Process         12.5.2       Bayesian Estimation for Parameters in the       Degradation Process         12.5.2       Bayesian Estimation for Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353     <		11.4	Experimental Studies	328
11.4.2       Case Study       332         References       335         12       RSL Prediction Approach for Systems with Operation State         Switches       337         12.1       Introduction       337         12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predict the RSL       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359       357         Part IV <th></th> <th></th> <th>11.4.1 Simulation Study</th> <th>329</th>			11.4.1 Simulation Study	329
References       335 <b>12 RSL Prediction Approach for Systems with Operation State</b> 337 <b>Switches</b> 337         12.1 Introduction       337         12.2 Literature Review       339         12.3 Problem Description for RSL Estimation       340         12.4 Model Formulation for Transitions Between the Operating       342         State and Storage State       342         12.4.1 Randomly Varying System Operation Process       344         12.5 Model Formulation of the System Degradation Process       344         12.5 Model Formulation of the System Degradation Process       344         12.5.1 Predicting the RSL Conditional on the Model       9         Parameters and Fixed System Operation Process       346         12.5.2 Bayesian Estimation for Parameters in the       9         Degradation Process       349         12.5.3 RSL Prediction Considering the Future Transitions       350         12.6 Case Study       353         12.6.2 Results and Discussions       357         References       359         Part IV Applications of Prognostic Information       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366 </th <th></th> <th></th> <th>11.4.2 Case Study</th> <th>332</th>			11.4.2 Case Study	332
12       RSL Prediction Approach for Systems with Operation State         Switches       337         12.1       Introduction       337         12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       342         12.4.2       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Operation Process       346         12.5.1       Predicting the RSL       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359       363         13.1       Introduction       363         13.2       Assumptions and Problem Description		Refere	ences	335
Switches       337         12.1       Introduction       337         12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       342         12.4.1       Randomly Varying System Operation Process       344         12.5       Model Formulation of the System Degradation Process to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       363         13.3       Introduction       363         13.4       Thorduction       363         13.2       Assumptions and Problem Description       366         13.2.1	12	RSL	Prediction Approach for Systems with Operation State	
12.1       Introduction       337         12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       342         12.4.2       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       363         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374		Swite	hes	337
12.2       Literature Review       339         12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State       342         12.4.1       Randomly Varying System Operation Process       342         12.4.2       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       367		12.1	Introduction	337
12.3       Problem Description for RSL Estimation       340         12.4       Model Formulation for Transitions Between the Operating State and Storage State.       342         12.4.1       Randomly Varying System Operation Process.       342         12.4.1       Randomly Varying System Operation Process.       344         12.5       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process.       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions.       357         References       359       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.2       Assumptions       367         13.3<		12.2	Literature Review	339
12.4       Model Formulation for Transitions Between the Operating State and Storage State.       342         12.4.1       Randomly Varying System Operation Process.       342         12.4.2       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process.       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions.       357         References.       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation		12.3	Problem Description for RSL Estimation	340
State and Storage State       342         12.4.1       Randomly Varying System Operation Process       342         12.4.2       Bayesian Estimation for Parameters in the System's       0peration Process         Operation Process       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estima		12.4	Model Formulation for Transitions Between the Operating	
12.4.1       Randomly Varying System Operation Process.       342         12.4.2       Bayesian Estimation for Parameters in the System's       344         12.5       Model Formulation of the System Degradation Process       344         12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process.       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions.       357         References       359         Part IV Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374			State and Storage State	342
12.4.2       Bayesian Estimation for Parameters in the System's Operation Process       344         12.5       Model Formulation of the System Degradation Process to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       363         13.2.1       Problem Description       363         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376			12.4.1 Randomly Varying System Operation Process	342
Operation Process       344         12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376			12.4.2 Bayesian Estimation for Parameters in the System's	
12.5       Model Formulation of the System Degradation Process       346         12.5.1       Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       Model Description       368         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       374         13.4.3       Estimating the RUL of PMS       377 <th></th> <th></th> <th>Operation Process</th> <th>344</th>			Operation Process	344
to Predict the RSL       346         12.5.1       Predicting the RSL Conditional on the Model         Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the       349         12.5.3       RSL Prediction Considering the Future Transitions       349         12.5.3       RSL Prediction Considering the Future Transitions       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2.1       Problem Description       366         13.2.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.3       Bayesian Updating of Model Parameters       376         13.4.3       Estimating the RUL of PMS       377 </th <th></th> <th>12.5</th> <th>Model Formulation of the System Degradation Process</th> <th></th>		12.5	Model Formulation of the System Degradation Process	
12.5.1       Predicting the RSL Conditional on the Model Parameters and Fixed System Operation Process.       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information         13       Reliability Estimation Approach for PMS       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376			to Predict the RSL	346
Parameters and Fixed System Operation Process       346         12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376			12.5.1 Predicting the RSL Conditional on the Model	
12.5.2       Bayesian Estimation for Parameters in the Degradation Process       349         12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376			Parameters and Fixed System Operation Process.	346
Degradation Process       349         12.5.3 RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6 Case Study       353         12.6.1 Background and Data Description       353         12.6.2 Results and Discussions       357         References       359         Part IV Applications of Prognostic Information       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.3 Estimating the RUL of PMS       377			12.5.2 Bayesian Estimation for Parameters in the	
12.5.3       RSL Prediction Considering the Future Transitions and Updated Parameters       350         12.6       Case Study       353         12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       363         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time.       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.3       Estimating the RUL of PMS       377			Degradation Process	349
and Updated Parameters       350         12.6 Case Study       353         12.6.1 Background and Data Description       353         12.6.2 Results and Discussions       357         References       359         Part IV Applications of Prognostic Information       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.2 Bayesian Updating of Model Parameters       376         13.4.3 Estimating the RUL of PMS       377			12.5.3 RSL Prediction Considering the Future Transitions	0.7
12.6 Case Study       353         12.6.1 Background and Data Description       353         12.6.2 Results and Discussions       357         References       359         Part IV Applications of Prognostic Information       363         13 Reliability Estimation Approach for PMS       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.2 Bayesian Updating of Model Parameters       376         13.4.3 Estimating the RUL of PMS       377			and Undated Parameters	350
12.6       12.6.1       Background and Data Description       353         12.6.2       Results and Discussions       357         References       359         Part IV       Applications of Prognostic Information         13       Reliability Estimation Approach for PMS       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.3       Estimating the RUL of PMS       377		12.6	Case Study	353
12.6.1       Background and Data Decemption       357         12.6.2       Results and Discussions.       357         References.       359         Part IV       Applications of Prognostic Information       363         13       Reliability Estimation Approach for PMS       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       376         13.4.3       Estimating the RUL of PMS       377		12.0	12.6.1 Background and Data Description	353
<b>Part IV</b> Applications of Prognostic Information       363 <b>13</b> Reliability Estimation Approach for PMS       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.2 Bayesian Updating of Model Parameters       376         13.4.3 Estimating the RUL of PMS       377			12.6.7 Results and Discussions	357
Part IV Applications of Prognostic Information         13 Reliability Estimation Approach for PMS       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time.       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.2 Bayesian Updating of Model Parameters       376         13.4.3 Estimating the RUL of PMS       377		Refer		359
Part IV       Applications of Prognostic Information         13       Reliability Estimation Approach for PMS       363         13.1       Introduction       363         13.2       Assumptions and Problem Description       366         13.2.1       Problem Description       366         13.2.2       Assumptions       367         13.3       Mission Process to Estimate the Mission Time       368         13.4       System Degradation Process to Estimate the Lifetime       374         13.4.1       Model Description       374         13.4.2       Bayesian Updating of Model Parameters       376         13.4.3       Estimating the RUL of PMS       377		Refere		557
13 Reliability Estimation Approach for PMS       363         13.1 Introduction       363         13.2 Assumptions and Problem Description       366         13.2.1 Problem Description       366         13.2.2 Assumptions       367         13.3 Mission Process to Estimate the Mission Time.       368         13.4 System Degradation Process to Estimate the Lifetime       374         13.4.1 Model Description       374         13.4.2 Bayesian Updating of Model Parameters       376         13.4.3 Estimating the RUL of PMS       377	Part	t IV	Applications of Prognostic Information	
13.1Introduction36313.2Assumptions and Problem Description36613.2.1Problem Description36613.2.2Assumptions36713.3Mission Process to Estimate the Mission Time36813.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description37413.4.2Bayesian Updating of Model Parameters37613.4.3Estimating the RUL of PMS377	13	Relia	bility Estimation Approach for PMS	363
13.2Assumptions and Problem Description36613.2.1Problem Description36613.2.2Assumptions36713.3Mission Process to Estimate the Mission Time.36813.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description37413.4.2Bayesian Updating of Model Parameters37613.4.3Estimating the RUL of PMS377		13.1	Introduction	363
13.2.1Problem Description36613.2.2Assumptions36713.3Mission Process to Estimate the Mission Time.36813.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description37413.4.2Bayesian Updating of Model Parameters37613.4.3Estimating the RUL of PMS377		13.2	Assumptions and Problem Description	366
13.2.2Assumptions36713.3Mission Process to Estimate the Mission Time.36813.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description.37413.4.2Bayesian Updating of Model Parameters.37613.4.3Estimating the RUL of PMS377			13.2.1 Problem Description	366
13.3Mission Process to Estimate the Mission Time.36813.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description.37413.4.2Bayesian Updating of Model Parameters.37613.4.3Estimating the RUL of PMS377			13.2.2 Assumptions	367
13.4System Degradation Process to Estimate the Lifetime37413.4.1Model Description37413.4.2Bayesian Updating of Model Parameters37613.4.3Estimating the RUL of PMS377		13.3	Mission Process to Estimate the Mission Time	368
13.4.1Model Description.37413.4.2Bayesian Updating of Model Parameters.37613.4.3Estimating the RUL of PMS377		13.4	System Degradation Process to Estimate the Lifetime	374
13.4.2Bayesian Updating of Model Parameters37613.4.3Estimating the RUL of PMS377			13.4.1 Model Description	374
13.4.3 Estimating the RUL of PMS			13.4.2 Bayesian Updating of Model Parameters	376
			13.4.3 Estimating the RUL of PMS	377
13.5 Reliability Estimation for PMS		13.5	Reliability Estimation for PMS	382

	13.6	Experimental Studies	383 383
	Refer	ences.	388 390
14	A Re	al-Time Variable Cost-Based Maintenance Model	393
	14.1	Introduction	393
	14.2	Degradation Modeling for Prognostics	395
		14.2.1 Degradation Modeling	395
		14.2.2 RUL Estimation.	398
	14.3	Replacement Decision Modeling.	399
	14.4	A Case Study	401
	Refer	ences	403
15	An A	daptive Spare Parts Demand Forecasting Method	
	Based	I on Degradation Modeling	405
	15.1	Introduction	405
	15.2	Degradation Modeling Description	407
	15.3	Adaptive Lifetime Estimation	408
	15.4	Adaptively Forecasting Spare Parts Demand	410
	15.5	Adaptive Parameter Estimation	412
	15.6	Case Study	413
	Refer	ences	416
16	Varia	ble Cost-Based Maintenance and Inventory Model	419
	16.1	Introduction	419
	16.2	Degradation Modeling for Prognostics	420
		16.2.1 Degradation Modeling	421
		16.2.2 RUL Estimation.	421
	16.3	Parameter Estimation	422
	16.4	Replacement and Inventory Decision Modeling	424
	16.5	Case Study	427
	Refer	ences	430

## Acronyms

AIC	Akaike information criterion
BM	Brownian motion
CBM	Condition-based maintenance
CDF	Cumulative distribution function
СМ	Condition monitoring
CTMC	Continuous-time Markov chain
EKF	Extended Kalman filter
EKS	Extended Kalman smoother
FHT	First hitting time
FPK	Fokker–Planck–Kolmogorov
FPT	First passage time
HMM	Hidden Markov model
HSMM	Hidden semi-Markov model
INS	Inertial navigation system
KF	Kalman filter
ME	Measurement error
MLE	Maximum likelihood estimation
MSE	Mean squared error
MTTF	Mean time to failure
PDF	Probability density function
PHM	Prognostics and health management
PMS	Phased-mission system
RE	Relative error
RSL	Residual storage life
RTS	Rauch-Tung-Striebel
RUL	Remaining useful life
STF	Strong tracking filter
TMSE	Total MSE

## Part I Introduction, Degradation Data Acquisition and Evaluation

#### Chapter 1 Advances in Data-Driven RUL Prognosis Techniques

#### 1.1 Introduction

Prognosis and health management (PHM) has drawn increasing attention and gained deepening recognition and widening applications during the past decades [1-4]. Actually, the initial health and usage inspection system was fist equipped in the early helicopters of US military and the synthetically health management philosophy was presented for spacecraft in the 1970s. Recently, the comprehensive solution for system performance prognosis and maintenance has been achieved in the Joint Strike Fighter F-35 project [5]. Further, the ability of PHM has already been listed by the Department of Defense (DOD) of United states as one of the essential norms for weapon system purchasing. This shows the significant implication of PHM in military fields. On the other side, industrial practice indicates that PHM technology can effectively reduce the maintenance cost, improve the reliability and guarantee the completion of tasks of the system [6, 7]. Research institutes including NASA [8], University of Maryland [6] and George's University [9], as well as commercial companies such as Boeing have launched a great deal of theoretical and applied research works about PHM technology. The PHM conference has been successfully organized and held by IEEE Reliability Society in Shenzhen, Macau, Beijing, Rome, and Zhangjiajie respectively, Beijing in six consecutive years since 2010.

Remaining useful life (RUL) estimation, offering guidance for sequential management involving inspection schedule, maintenance, replacement and spare parts ordering, has been considered as the kernel technology of PHM, and the focus of current research in the field of reliability also. According to Petch's classical monograph about PHM technology [6], methods for RUL estimation can be classified into three kinds: namely physical model-based methods, data-driven methods and their combinations. However, with the development of industry and the continuing extension of human exploring activities, the complexity of a system, together with the diversity and uncertainty of its operating environments, continues to increase, which results in extreme difficulties in constructing physical models capturing the system and its operating circumvents. Meanwhile, data-driven methods, including

Springer Series in Reliability Engineering, DOI 10.1007/978-3-662-54030-5\_1

<sup>©</sup> National Defense Industry Press and Springer-Verlag GmbH Germany 2017

X.-S. Si et al., Data-Driven Remaining Useful Life Prognosis Techniques,

artificial intelligence-based methods and statistical data-based methods have become an effective avenue to evaluate reliability and estimate RUL, especially for vital systems with high reliability and long lifetime. Artificial intelligence based methods can hardly provide a probability density function (PDF) estimate capturing stochastic and uncertain characteristics of the RUL, while this desire is a natural result for stochastic data-driven methods [10]. To address the uncertainty of prognosis, we mean statistic data-driven methods as data-driven methods throughout this chapter. According to the observability of underlying degradation process, Si et al. provided an review on data-driven methods for both direct and indirect observed degradation data, introducing many common methods including Gamma processes, Wiener processes, Hidden (semi-) Markov models, stochastic regression models, stochastic filtering-based models and covariate hazard-based models, from the perspectives of applying procedure, merits and drawbacks [11]. While being satisfactory for RUL estimation under each specific applying condition, these methods exhibit some limits in cases with heterogeneity from the inner states or the external operating conditions of systems.

Heterogeneity is widespread in the inner states of the system and the related working environments. Examples involve that a weapon system may experience various operating conditions, saying storage, inspection, transport, and maintenance during its life cycle due to different tasks; that a manufacture system produces different products under different workloads; and that even systems from the same category may exhibit various degrading paths in the same environment. The performance degradation of a system is a result of interactions of both inner deterioration and working environment of the system, indicating a need for incorporating the heterogeneity into degradation modeling, to achieve a more accurate RUL estimation. For particular heterogeneity, such as the unit-to-unit variability, changing working conditions and periodic tasks, many recent advances in RUL estimation have appeared. However, to the best of the authors' knowledge, there is still no review regarding degradation modeling and RUL estimation for systems with heterogeneity. Therefore, this chapter tries the best to fill this gap.

Toward the end of this chapter, three kinds of heterogeneity are considered consecutively: the unit-to-unit variability for systems from the same category, the variability in time-varying operating conditions, and the diversity of tasks and workloads of system during their life cycles. The first kind of variability describes the differences in degradation processes of units from the same category, while the second represents noninform working conditions related to the degradation, such as the time-varying, multi-state and stochastic working environments or random shocks. The third kind of heterogeneity captures the influence of changes in tasks and management activities involving inspection, maintenance, etc. Accordingly, this chapter classifies methods addressing degradation modeling and RUL estimation with heterogeneity into three kinds, each of which considers one kind of heterogeneity introduced above and consists of some subclassifications. The taxonomy of RUL estimation approaches for systems with heterogeneity is illustrated in Fig. 1.1.

The remainder of the chapter is structured as follows: Section 1.2 summarizes methods considering unit-to-unit variability, saying degradation models with random



Fig. 1.1 Taxonomy of RUL estimation approaches for system under heterogeneity

effect In Sect. 1.3 methods considering the impact of the working conditions are provided. Methods for incorporating the influence of diversity in tasks and workloads are reviewed in Sect. 1.4. Section 1.5 concludes the chapter and provides several possible directions for future studies.

#### **1.2 Methods Considering Unit-to-Unit Variability**

A large number of experiments and engineering phenomena show that systems of the same category, even from one batch degrade differently from one another in performance. This kind of difference in degradation is usually defined as the unit-to-unit variability, due to the variability in inner structures of the considered systems, as well as the diversity in their working environment. Commonly, models with random effects are employed to capture the unit-to-unit variability, when we model the degradation process and estimate the RUL. The most typical way to do so is to specify some parameters of the model as random variables governed by distributions with computing convenience, presenting the individuality in degradation processes from different units and leave the rest of parameters as constants describing the universality in degradation of systems from the same category or batch. In the following, random coefficients regression models and stochastic process models with random coefficients of this kind are discussed, respectively.

#### 1.2.1 Random Coefficients Regression Models

Random effects first appeared in random coefficients regression models. In the most frequently cited paper about degradation modeling and RUL estimate [12], Lu and Meeker described the random coefficients regression model in a general form as

$$X(t_{ij}) = g(t_{ij}; \boldsymbol{\phi}, \boldsymbol{\theta}) + \varepsilon_{ij}$$

where  $X(t_{ij})$  is the amount of the degradation of the *i*th device in the *j*th inspecting time  $t_{ij}$ , the fixed coefficients  $\phi$  and the random coefficients  $\theta$  are, respectively, used to characterize the universality and individuality in degradation of different systems, and is the random noise.

Random coefficients regression models have been extended, developed and applied widely in many areas, in which a series of extended works presented by Gebraeel et al. are included [13–15] Son et al. compared various kinds of RUL estimation method based on random coefficients regression models [16]. Suk and Paul proposed a nonlinear random coefficients regression method for degradation data [17], and applied the model to the degradation of the vacuum fluorescent tube display. To improve the accuracy of parameter estimation, Weaver and Meeker also studied the optimal design of repeated measures degradation studies, and the method to design accelerated repeated degradation studies [18, 19]. A procedure deciding the minimum sample size and the minimum times of systematic sampling for each item to achieve an anticipant accuracy of estimation (large sample approximate variance) has been provided in their works.

However, according to Wang's analysis in [20], the assumptions of random coefficient regression models result in several limitations, involving the need for more historical degradation data from different systems of the same category, the difficulty in capturing the time-varying dynamics of systems and the independency between random noise with time.

#### 1.2.2 Stochastic Process Models with Random Coefficients

Incorporating random coefficients into stochastic degradation process-based models enables both considerations of time-varying dynamics of an individual system, and description of unit-to-unit variability, and thus has been favored by many researchers. Suppose that the degradation of a system is modeled by a stochastic process  $\{X(t); t \leq 0, \theta, \vartheta\}$ , with constant parameters  $\theta$  and random parameters  $\vartheta$ . Under the concept of first passage time (FPT), the RUL of the system conditional on the observation  $X(t_k)$  at time  $t_k$  is defined as

$$L_k := \inf \{l_k : X(t_k + l_k) \ge \omega | X(t_k) < \omega \},$$

where is a preset constant failure threshold.

Lawless proposed a Gamma process-based model containing the covariates and random effects, and applied it to degradation modeling and RUL estimation [21]. When fitting the semi-parametric Gamma process to degradation data, Ye et al. also took the random effects into consideration. Further, the unit-to-unit variability was captured using random parameters following some particular distributions in recent degradation models based on Inverse Gaussian process [22, 23]. The same specifications addressing differences in the degradation process of systems from the same category were used in the application of Inverse Gaussian process for systems with monotonous degradation by Wang [24] and Ye et al. [25]. For nonmonotonic degradation processes with fluctuations, Wang proposed a Wiener degradation model with random effects [26]. Si et al. presented a degradation path-based RUL estimation method with exact closed form solution of the estimated PDF of the RUL in linear and exponential cases, which also incorporated the random effects. Peng and Zeng analyzed the misspecification of linear degradation model in the framework of Wiener process with random drift coefficient [26]. Similarly, Si et al. [27] and Wang et al. [28] set some parameters in their methods as normally distributed random variables, when modeling nonlinear diffusion degradation process and additive hybrid degradation process, respectively. From results in the existing literature, stochastic process models with random effects can effectively improve the estimation accuracy and extend the applications of the initial degradation models, in both cases of monotonous and nonmonotonous degradation processes no matter linear or nonlinear.

In the industrial applications, the main flaw of degradation models with random effects is the complexity in computation. Therefore, the primary concern choosing the random parameters and their distribution models is the convenience of calculation. Normally distributed random variables are with high frequency in the related literature. For example, in Tseng and Yu [29], Lu and Meeker [12], Gebraeel [13], Si [30], all selected models with random variables following Normal distribution to characterize unit-to-unit variability. As for some particular degradation models, parameters subjected to special forms of distributions are preferred. Wang utilized Gamma distributions to mode the drift and diffusion coefficients in the Wiener degradation model [26], and Ye el al. also used gamma distributed parameters when constructing semi-parametric Gamma degradation process. These choices are made due to the purposes of computing convenience. The misspecifications of such distributions are considered by some researchers and some nonparametric distributions based on observations are recommended [31-34]. However, explicit results of the estimated RUL can hardly be derived when nonparametric distributions are used. Besides, the corresponding computation is always complicated, which makes it inadequate for real-time RUL estimation. Therefore, it is a challenge to reasonably choose random parameters and their distributions that cannot only capture the unit-to-unit variability but also benefit computation, when using degradation models with random effects.

#### **1.3 Methods Considering Impact of Heterogeneity** in Working Environment

#### 1.3.1 Methods Based on Stochastic Filtering

As early as 1979, Sarma et al. estimated health state of aerospace engine using Kalman Filter (KF) technology, and achieved a maintenance decision optimization based on the estimated results [35]. Afterwards, Wang and Christer [], Batzel and Swanson [], proposed different state evaluation and RUL estimation methods, applied successfully to electromagnetic induction smelting furnaces, aeronautical batteries and other industrial systems, based on the construction of state-space models. As for the nonlinear and nonGaussian state-space models, Extended KF, Benes Filter, Multiple Model Filter and Particle Filter based methods for health state and RUL estimation, have been successively proposed [36–40]. When the impact of heterogeneity is incorporated into stochastic filter-based methods, two kinds of sub methods can be referred to, namely semi-stochastic filter based methods and adaptive parameter based methods.

Ability to handle unobservable degradation is an advantage of stochastic filter based methods, while the failure threshold of the unobservable degradation can hardly be specified. In this connection, the lifetime of a system is directly defined as a state in the state-space model by Wang and Christ, and the length of time interval between two consecutive inspections is treated as the decrease of the lifetime. As such, RUL estimation method based on stochastic filter was proposed in [20] through constructing a stochastic relationship between the condition monitoring data and the lifetime of the system. This original method has been extended to cases where the operating environments are considered, by establishing the stochastic relationship between lifetime with the condition monitoring information and the operating environments simultaneously [41].

In another class of approaches for degradation modeling, some important parameters are expanded as state of a state-space model, which is utilized to describe dynamics in parameters. These parameters are adaptive to the changing environmental variables and updated jointly online with the healthy state of the system. As a result, the updated states and parameters are substituted to obtain a new estimation of RUL. Wang and Mattgew set the drift coefficient in Wiener process as an adaptive parameter, which will be updated through KF technology once new observations are available [42]. Inspired by [42], Si et al. proposed a Wiener degradation model with nonlinear drift coefficient function, which also makes some parameter adaptive to the observed data [43, 44]. In this chapter concerning models for RUL estimation under three sources of variability, the drift coefficient was also treated as an adaptive parameter and expanded to a state in the state-space model describing the degradation of the system, and was updated with the degradation level on-line [45]. The successful application of this method in the RUL estimation of an inertial navigation system has shown much superiority of such method. The best advantage of such stochastic filter based RUL estimation methods lies in that the parameters and the accordingly RUL estimation can be updated with the newly observed condition monitoring information. Furthermore, the dynamics in the degradation process and the probable measurement errors are taken into consideration, which makes it suitable for indirectly observable degradation process. However, these methods have a premise in common that an explicit state-space model must be constructed, which may be impossible in some cases. Another limitation is that the RUL estimation is obtained without consideration of the possible future changes in the degradation. In addition, the assumption in the semi-stochastic filter based method that there is a deterministic equal relationship between the reduction of lifetime and the inspection interval may not hold in many cases, especially when there are changes in the operating environments of the workloads of the system.

#### 1.3.2 Multi-stage Degradation Models

Multi-stage degradation models are proposed to handle the period differences existing in the degradation process. In Wang's two stage degradation model, the degradation data after the defect point were used to estimate the parameters in the degradation model and predict the RUL of the system [46]. In order to evaluate the remaining storage lifetime of a system, Feng el al. proposed a multi-stage Wiener degradation model in [47], where some related works were listed. These works include the nonhomogeneous Poisson process which can be used to analyse time-varying failure rate of software, the nonlinear model with random coefficients which is applied to the multi-state nonmonotonic degradation process of hardware, the multi-stage linear regression model, the multi-stage linear stochastic process model, and so on. Li and Pham studied the reliability modeling problem of multi-state degrading systems, under the interaction of multi competing failure modes and random shocks [48]. The common shared by these models is the presence of change points, such as the defect point in two-stage model and the starting/finish points of each stage. Generally, the unknown locations of these change points have to be determined by selecting appropriate detection methods before model identification and RUL estimation. This problem has been considered as highlight but also aporia. Currently, the maximum likelihood estimation, stochastic filtering, and control charts are the most popular methods to estimate the change points in multi-stage degradation models. Thus, the accuracy of change points estimate has direct influence on the accuracy of the RUL estimation. Another popular multi-stage degradation process for RUL estimation is the Markovian model. To model the hidden degradation process, Hidden Markovian Model (HMM) was first introduced to RUL estimation and condition based maintenance (CBM) [49]. On this basis, Dong et al. proposed the RUL estimation framework by using Hidden semi-Markov Models (HsMM), which extends the exponential assumption of state sojourn time to more general situations [50]. He and Dong extended the work in [50] and obtained RUL estimation through a comprehensive consideration of sojourn time in each state which has been modeled

by a single HsMM [51]. Prognosis of both performance and RUL were achieved in [52] by a combination of HsMM and AR model for time series data. Giorgio, Guida and Pulcini considered the age- and state-dependency of the degradation process in the framework of Markovian degradation model [53, 54]. A very good result was obtained when the proposed models were applied to the degradation process of marine engine cylinder.

The proposition of these models improves the accuracy of degradation modeling, and enriches the selections of models for different degradation processes. However, almost all multi-stage degradation models face the problem of determining the number of degradation stages, and a large amount of training data as well as a complex computation procedure are needed for parameter estimation. Further, instead of a derivable analytical solution to the PDF estimate of the RUL, a time-consuming simulation based methods have to be executed. In addition, the RUL estimation in multi-stage degradation models is based on the information since the latest change point. Such an estimation of RUL is accurate if there will be no change occur in the future time of the system. In more practical situations with possible change points in the future, severe bias will be introduced into the RUL estimation if using such estimation mechanism, i.e., ignoring the possible change points in the future. To tackle this problem, the possible change points in the future should be considered in multi-stage degradation processing modeling and RUL estimation.

#### **1.3.3** Covariate Hazards Model

Factors that affect degradation in performance of systems are defined as covariates in engineering pactive. The classical model for lifetime analysis, named proportion hazards model, is the most widely used ones in the fields of RUL estimation, reliability analysis/evaluation, decision-making and optimization on maintenances, etc. The existing works related to hazards models have been reviewed in [55]. The description of the system failure rate is the core of the proportional hazards model, and also the key for reliability assessment and RUL estimation. Failure rate in the proportional hazards model usually consists of the product of a reference failure rate function  $h_0(t)$  and the covariate function  $\psi(\beta z(t))$ ,

$$h(t|\mathbf{z}(t)) = h_0(t)\psi(\boldsymbol{\beta}\mathbf{z}(t)),$$

where z(t) are the covariate variable,  $\beta$  are the regression coefficients which can be estimated using historical lifetime data or censored lifetime data of the system from the same category. Proportional intensities model and proportional covariates model, developed from the proportional hazards model, are also popular models for RUL estimation [].

In heterogeneous working environments, the failure rate of the system will be affected. Ye et al. studied the influence of heterogeneity in the working conditions on the estimation of the RUL, based on the analysis of accelerated life test [56].

A system may experience fixed, time-varying, and even stochastic environmental conditions and the corresponding covariates may also be constants, time-varying or even random variables. In order to characterize the influence of random covariates on the failure rates in the proportional hazard models, researchers have considered using some stochastic process to model the time-varying covariates, and incorporating the modeled covariates into the proportion hazard models. For example, Markov chains, which can naturally describe the operating process of a system, are the most frequently used process to model the changing procedure of covariates [57–59]. A HMM with a known state transition law was utilized to model the stochastic degradation process, and the formula to calculate the mean RUL was derived in [60]. Lu and Liu studied the relationship between failure rates and the dynamic working environment [61]. In their research, the changing covariates were modeled by a two states (normal/severe) Markov Chain, as such, failure rate functions are changing with operation function, and the lifetime of the system can then be determined by its working conditions.

Although their strong explanatory property makes covariate hazard models dominant both in theory and application researches, this kind of method does have some inborn limitations, which have been summarized by Si et al. in [11]. Furthermore, some difficulties should be solved before the practical applications of this kind of methods. First, with the development of high reliability and small amount systems, the lifetime data required for estimating parameters  $\beta$  and reference failure rate function  $h_0(t)$  are difficult and expensive to obtain. Second, it is hard to determine the form of covariate function when systems become complex.

#### 1.3.4 Degradation Models Involving Random Shocks

During the degradation process, a system may suffer various kinds of shocks, which will impact the performance of the system as well as its underlying degradation process. Typically, there are five different types of random shock models existing in the literature [62]: (i) extreme shock model: the system fails when the size of a shock is beyond a specified threshold value; (ii) cumulative shocks model: a system fails when the accumulated damage of shocks is beyond a critical level; (iii) *m*-shock model: a system failures after suffering *m* shocks whose sizes are greater than a critical level; (iv) run shock model: failure occurs when there is a series of n consecutive shocks that are greater than a threshold; and (v) shock model: a system experiences failure when the inter-arrival time of two sequential shocks is less than a threshold. As for reliability modeling and RUL estimation under random shocks, there have been a number of studies including [63, 64] to which we can refer. Random shocks, whose influences on the performance of a system are addressed in this section, are regarded as heterogeneity in the working environment. In general, failure is a result from the interaction and competition of the performance degradation and external random shocks [65–67]. Poison process (homogeneous/nonhomogeneous) [68], Markov Chains [69], and the phase-type distribution [70, 71] have all been used to describe the arriving process of random shocks. Models for degradation processes with random shocks can be divided into two categories here, according to the existing of interdependency between the continuous degradation processes with random shocks.

Degradation processes and random shocks are supposed to be independent of the first type of models. Klutke and Yang first proposed an availability model for the system under interaction of degradation and random shocks [65]. Afterward, Huang and Askin analyzed and constructed a reliability model for systems under the competing impact of the degradation process and sudden failure [72]. Li and Pham proposed a reliability model for a system suffering two types of degradation and a type of random shocks [73]. Chen and Li assumed that from the external environment the degrading system may experience two types of random shocks, i.e., fatal or nonfatal [74]. An optimal maintenance strategy was proposed under a further assumption that system's tolerance of the total number of nonfatal shocks decreases subjected to the times of maintenance. A common underlying assumption of the works in [65, 72–74] is that degradation processes causing softer failures and random shocks leading to hard failures are independent from each other, and no mutual influence between degradation and shocks exists.

The interactions between shocks and degradations are considered in the second type of models. When studying the reliability and maintenance model for the system under competing degradation process and random shocks, Wang and Pham supposed that fatal shocks caused a direct failure of the system while nonfatal shocks resulted in abrupt increases in the degradation level [75]. The interdependency of soft failure caused by degradation and the hard failure caused by random shocks was included in Peng's work about reliability modeling [76]. Liu et al. considered the relationships between the failure rate of a system with age, degradation level and their interactions in the degradation model [64]. Recently, Koosha studied the influence of various types of random shocks on the degradation processes of the system and supposed that the level of degradation process jumped once a shock came while the degradation rate changed only after a particular type random shock [62]. A reliability model for dependent competing failure processes with changing degradation rate was then proposed based on this dependency of degradation process on random shocks.

The primary drawbacks of using such kind of methods are the following: (1) Lots of existing works incorporated random shocks to the linear degradation process, while degradation processes of actual systems are often nonlinear. To be more practical, the influence of random shocks on nonlinear degradation processes should be considered, which has seldom been done except [76, 77]. (2) As for discretely inspected system, the time and influence amplitude of random shocks can hardly be measured directly, which may introduce extreme difficulties in model identification and parameter estimation. (3) In cases where the dependency between degradation processes with random shocks is considered, attention has been focused on the influence of random rocks on (levels and rates) degradation process, while researches about impact of degradation on random shocks and the interdependency between each other are rarely reported except for [77, 78]. (4) Random shocks in degradation level and

even the failure of the system. However, there exist some shocks improving the system's performance, e.g., the state-of-health regeneration phenomena in lithium-ion battery systems. As such, this kind of shock should be further considered into the degradation process in the future.

# **1.4** Methods Considering the Impact of Tasks and Workloads

# 1.4.1 Degradation Modeling for Systems with Dynamic Workloads

Due to the diversity and randomness in the operating environments and workloads of systems, the characteristic of degradation varies with age throughout the whole lifetime cycle. If the dynamic operating model of the system's workload is constructed scientifically and incorporated into the degradation model reasonably, a more accurate estimation of the RUL will be achieved.

During the industrial applications, some systems change their working state in several different working modes, corresponding to which are the different workloads and various degradation processes. For example, a missile weapon system with an extremely long storage before being launched may experience different working states involving storage, transportation, inspection and maintenance during its service. Studies have shown that, due to influences of temperature, humidity and human factors in the storage conditions, the performance of gyroscopes installed in an inertial navigation system (INS) exhibit some decreasing trends, which will be accelerated by each electrifying inspection, after some time of storage [79]. Moreover, the states switching of the system is a stochastic process, because of the uncertainty in the coming of different tasks or missions A continuous-time Markov model (CTMM) with finite state is a natural selection to describe such a stochastic operating process [80, 81]. In literature, CTMM was used to capture the time-varying random working conditions of a system in Jeffrey and Steven's stochastic models for degradationbased reliability [82]. Si el al. also utilized two-state CTMM to represent the states switching process between storage and usage, and the operating model was successfully applied to estimate the remaining storage life (RSL) of gyroscopes in INS [83]. Hawkes proposed a reliability assessment model based on the CTMM modeling of working condition switches. Huynh modeled changing working conditions using CTMM and incorporated the results into the decision-making framework for adaptive CBM decisions [84]. Another focus when the dynamics of the workloads are of concern is to establish the relationship between the operating conditions and the degradation process of the system. This relationship is usually supposed to be totally known or at least particular functions with unknown parameters which can be estimated by using the observations of both operating conditions and degradation process. Jeffery pointed out that this relationship varies case-by-vase and should be

determined according to the specific characteristics of the device [82]. When Wiener process was used as the degradation model, Si el al. assigned different drift coefficient values for the system in the state of usage and storage, respectively [83]. Besides, Arrhenius model and Eying model are frequently used to represent the relation between the degradation and working environments for electromechanical systems.

It is worth noting that the existing methods suffer some limitations. On one hand, the primary limit of CTMM is that the sojourn time in each state is exponentially distributed, which may be incompatible with the facts in practical applications in industry. To overcome this shortcoming, a semi-Markov model can be employed. Besides, when the operating information cannot be recorded directly, the according HMM and HsMM should be used to model the operation process of the system. On the other hand, with more and more complex structures of systems, the relationship between the degradation with operating conditions can be neither characterized by the simple existing laws, nor constructed through physical analysis, which may restrict the application of this kind of methods.

#### 1.4.2 Degradation Modeling for System with Maintenances

Maintenance is an effective way to remove faults, reduce failure rates and improve the reliability throughout the lifetime of the system. Scientific and reasonable maintenance schedule can efficiently reduce the operating costs and the risks of the system, which also works for degrading systems [85]. Degradation modeling and maintenance activities of systems are closely related. On the one hand, the results of reliability evaluation and RUL estimation based on the degradation data offer the health evaluation information required for scheduling maintenance activities. On the other hand, maintenances improve the performance of the system and thus change the degradation path. To extend the application of the degradation model and improve the accuracy of RUL estimation, the influence of maintenance on degradation should be taken into consideration.

There are plenty of studies addressing preventive maintenance and optimal inspection based on degradation modeling [86–88], and some relate to the effects of maintenances on systems' performance [89, 90]. Popular models include the 'repair as new' model and the 'repair as old' model [91]. Both kinds of models assume that the performance of the system will be improved by maintenances, and the hazards functions are used to describe the effects of maintenance activities. In the 'repair as new' model, the system can be restored to the original state after a perfect maintenance, usually corresponding to the hazard increasing model [92]. The 'repair as old' models assume that maintenances on a system are imperfect so that the performance of the system recovers to a level worse than that of new system [93] Besides, the system degrades until the next maintenance or failure whichever comes first. Recently, Wang et al. employed a renewal-reward process perturbed by a diffusion, which is also defined as a Wiener process with random jumps elsewhere, to model